Author's response to reviews

Title: Assessing feasibility and acceptability of study procedures: getting ready for implementation of national stroke guidelines in out-patient health care

Authors:

Susanne Palmcrantz (susanne.palmcrantz@ki.se)
Malin Tistad (malin.tistad@ki.se)
Ann Catrine Eldh (ace@du.se)
Lotta Widén Holmqvist (lotta.holmqvist@ki.se)
Anna Ehrenberg (aeh@du.se)
Göran Tomson (goran.tomson@ki.se)
Christina B Olsson (christina.b.olsson@sll.se)
Lars Wallin (lwa@du.se)

Version: 4
Date: 5 November 2015

Author's response to reviews: see over
We wish to thank the reviewers for a thorough review and wise suggestions. Please see the comment from reviewer 2 and our reply below.

Title: Assessing feasibility and acceptability of study procedures: getting ready for implementation of national stroke guidelines in out-patient health care
Version: 3
Date: 30 September 2015
Reviewer: Emma Power
Reviewer's report:
Discretionary issues/revisions
My major concern still remains the use of a qualitative analysis system that doesn't use the specific data in terms of quotes to support the thematic and content-based findings. Even if this had one quote for each item in the table (table 3) it would help show the work and where it came from without significant word length issues?

Reply from the authors: We have now added quotes to support the findings presented in Table 3.